Author's response to reviews

Title: Benign cystic mesothelioma of the appendix: a case report

Authors:

Donal B O' Connor (donaloconor@yahoo.com)
David Beddy (davbeddy@rcsi.ie)
Muyiwa A Aremu (natiestefmel@gmail.com)

Version: 5 Date: 9 October 2010

Author's response to reviews: see over
Dear Editor,

We have revised to manuscript as requested. We have made all changes as outlined below.

Major revisions (we require the author to make these changes)

References: Please make sure references are cited in numerical order. We notice reference 10 is not cited within the main text. We also notice there is a reference 15 cited in the text. As the reference list only goes up to 10, please amend this discrepancy.

Reference 15 has been changed to ref 10 as 15 was an error.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, `The Title of My Prestigious Publication: A Scientific Study.' should be changed to `The title of my prestigious publication: a scientific study'

Changed on manuscript and submission system

- Title: please amend your title to include `: a case report' on the submission system
- Email addresses: Please use authors’ initials to indicate affiliation to email addresses
- Tracking/coloured text: Please remove all tracking or coloured text from the manuscript
- Structure: Please change ‘Case report’ to ‘Case presentation’
- Authors’ contributions: Please format this section into a paragraph
- Competing interests: Please move this section to immediately after the Conclusions section
- Additional files: Please remove the cover letter from the additional files

Yours sincerely,

Donal O’ Connor